# Do Elite Schools Benefit the Average Student?

János Divényi Sándor Sóvágó\*

#### Abstract

This paper studies the effects of enrollment in an elite school on elite-school students' academic achievement in Hungary. Enrollment in a Hungarian elite school entails having academically stronger peers and early switching to a secondary school. We examine effects for elite-school students throughout the outcome distribution using a mild stochastic dominance assumption. We find that enrollment in an elite school decreases female and low-ability students' mathematics test scores two years after enrollment. However, these negative effects are short-lived, and we obtain estimates that are consistent with substantial positive effects four years after enrollment. School value-added estimates lie within our non-parametric bounds, and confirm the positive effects on the medium run.

**Keywords**: elite schools, partial identification, school effectiveness, selective education.

JEL-codes: C21, I21, I24, I26.

<sup>\*</sup>This version: September, 2020. Divényi: Central European University, Hungary (e-mail: divenyi\_janos@phd.ceu.edu), Sóvágó: University of Groningen, Netherlands (e-mail: s.sovago@rug.nl). The authors thank Gábor Békés, Gábor Kézdi, Erik Plug, Anthony Strittmatter, Bas van der Klaauw, and the seminar participants in Amsterdam and Budapest for their comments and suggestions.

# 1 Introduction

Most school districts feature *elite schools* in which admission is merit-based (i.e., depends on a qualifying priority score) and the student body consists of affluent, high-achieving students. Admission to these elite schools is highly selective and preparation for the admission requires substantial pecuniary and non-pecuniary resources from the parents of prospective students. Therefore, it is important for these parents to understand the extent to which their child may benefit from enrollment in an elite school.

A large literature studies the effectiveness of elite schools (for a recent summary, see Beuermann and Jackson, Forthcoming). This literature exploits discontinuities in admission chances that are created by the merit-based priority-score cutoffs. Thereby, these studies provide estimates for students who are on the margin of admission. However, if the effect of elite schools is heterogeneous, existing estimates may not be informative for parents, who would be interested in the effect for the *average* elite-school student with similar characteristics to theirs.

This paper studies the effects of elite schools on academic achievement in Hungary. Meritbased admission combined with high demand implies that elite-school students have highachieving peers. Elite schools offer a more advanced, higher-paced curriculum, and teachers of high-qualification. Moreover, elite-school enrollment entails early switching to a secondary grammar school. Using administrative data, our study examines how enrollment in an elite school affects elite-school students' short- and medium-run academic achievement. Motivated by the potential heterogeneity in the effectiveness of elite schools, we conduct our analysis by gender and baseline ability (Oosterbeek et al., 2020).

Our main empirical strategy identifies the effects of enrollment in an elite school using non-parametric bounds. Our non-parametric bounds approach builds on a weak stochastic dominance assumption. We assume that conditional on observable student characteristics more able students are more likely to enroll in an elite school (conditional Monotone Treatment Selection, MTS; see Manski and Pepper, 2000; de Haan, 2011). The conditional MTS assumption yields an upper bound on the *average treatment effect on the treated* (ATET). We estimate the effect of elite-school enrollment throughout the outcome distribution, thus, we can study whether the effects differ between the top and the bottom end of the outcome

distribution (de Haan and Leuven, 2020).

The non-parametric bounds approach offers several advantages. First, our approach identifies the effect of elite-school enrollment for elite-school students, and not only for students on the margin of admission. Second, the conditional MTS assumption is consistent with elite schools' admission policy and yields testable implications.

We find that enrollment in an elite school has a negative effect on female and low-ability elite-school students' mathematics test scores two years after enrollment. Elite-school enrollment reduces the probability of scoring above the median in mathematics by more than 7.5 percentage points, corresponding to 30 percent, for low-ability female students, and by more than 1 percentage points, corresponding to 10 percent, for low-ability male students. For high-ability female students, we find that elite-school enrollment reduces the probability of scoring in the top 10 percent in mathematics by more than 2 percentage points, corresponding to 10 percentage points, corresponding to 10 percent. We argue that our findings for female and low-ability students are not the consequence of grading policies (e.g., ceiling effects, grading on a curve), but they reflect negative effects on skill formation. By contrast, we cannot rule out positive effects for high-ability male students at the upper half of the outcome distribution.

We next investigate whether the negative short-run effects persist four years after enrollment. We find that the upper bounds are positive and relatively large in magnitude for each ability and gender group. Thus, our non-parametric bounds strategy is uninformative about the sign of the effect on the medium run.

We further examine the medium-run effects of elite-school enrollment using a complementary empirical strategy. Using the selection on observables assumption, we estimate school value-added models to identify the average treatment effect on the treated. Our school value-added models control for students' lagged test scores and well as a rich set of student and school characteristics. We find that elite-school enrollment has a positive effect eliteschool students' mathematics and reading test scores on the medium run. Even if the school value-added and non-parametric bounds empirical strategies build on different identifying assumptions, reassuringly, the school value-added estimates are always consistent with our non-parametric bounds on the effect of elite schools.

Our paper contributes to the understanding of the effectiveness of elite schools in a number of ways. First, numerous studies examine the effects of elite schools (or attending a better school) using regression discontinuity design (RDD) in settings where admission is merit-based (Jackson, 2010; Clark, 2010; Pop-Eleches and Urquiola, 2013; Bui et al., 2014; Lucas and Mbiti, 2014; Abdulkadiroglu et al., 2014; Dobbie and Fryer, 2014; Anderson et al., 2016; Barrow et al., Forthcoming).<sup>1</sup> These RDD estimates are informative for students who are on the margin of admission. Instead, our study focuses on the effect for the average student, and thus complements this literature. Moreover, these RDD studies have limited ability to identify heterogeneous effects by student ability, since student ability varies little at the margin of admission (Oosterbeek et al., 2020). By contrast, our identification strategy allows us to estimate the effect for different ability groups. When we split the sample by baseline ability, our estimates suggest that the benefits of elite schools are concentrated on high-ability students. When looking at the outcome distribution, our estimates indicate that the benefits of elite schools materialize at the top of the outcome distribution.

Second, our paper studies the effect of elite-school enrollment on outcomes that are measured in different points in time. Documenting how the effects of elite school enrollment change over time is important, since behavioral responses of students, teachers, or parents may materialize on different time horizons (Pop-Eleches and Urquiola, 2013). We show that elite-school enrollment has a negative effect on female and low-achieving students' mathematics score on the short run. This finding suggests that it more costly for certain groups of students to adjust to the elite-school environment.

Finally, we also contribute to the studies evaluating the performance of Hungarian elite schools. In the same context as ours, Horn (2013) finds positive, albeit, imprecisely estimated effects of selective secondary-school attendance on short-run academic achievement. Using less stringent identifying assumptions, our study considerably narrows the upper bound on these estimates.

The remainder of the paper is organized as follows. Section 2 provides background information about Hungarian education and the organization of elite schools. Section 3 describes

<sup>&</sup>lt;sup>1</sup>A very much related literature studies elite-school effectiveness (or the effects of attending a better school) in settings where a lottery-based admission system is in place (e.g., Cullen et al., 2006; Deming, 2011; Dobbie and Fryer, 2011; Oosterbeek et al., 2020). Our study is similar to these papers in a sense that it identifies the effect of elite-school enrollment away from the priority-score cutoff. A key difference is that admission is merit-based in our setting, which implies that elite-school students are more selected that those who are admitted in a lottery-based admission system.

our data and provides summary statistics. Section 4 discusses our empirical strategies and presents the results of our validity checks. Section 5 presents our results. Section 6 concludes.

# 2 Context: Elite schools in Hungary

This section describes the institutional details of elite schools in Hungary. We describe admission to elite schools and detail the treatment.

Students begin primary education at age 6 in Hungary. After eight years in primary schools, they transition to secondary education. Secondary education is tracked in Hungary: students may choose a secondary grammar school, which has a more academic orientation and prepares students for higher education, or students may choose a vocational school, which has a less academic orientation and gives a vocational degree (see Table 1). Primary and secondary education are organized together in some schools. That is, in these *comprehensive schools*, students do not switch school when they transition from primary to secondary education. Even in these comprehensive schools, students are required to do the admission exam to proceed on the secondary track.

Primary-school students, at the end of grade 6, may decide to enroll in the 6-years long academic track of a secondary grammar school. We label these 6-years long academic tracks as *elite schools*. These elite schools are typically separate classes in a secondary grammar school, which have regular tracks as well. Thus, an elite school is essentially an "elite track" in a secondary grammar school (cf. Pop-Eleches and Urquiola, 2013).<sup>2</sup>

Admission to elite schools is merit-based. Elite schools organize admission exams where students have to solve numeracy and literacy tests. While all applicants solve the same test in the country, the priority-score formula used for the actual rankings is school-specific: it is a combination of the students' primary-school grades, the result of their written admission test, and the result of their oral admission exam. Since elite schools are highly oversubscribed, they are highly selective, and only high-achieving students are admitted.

Enrollment in an elite school is a composite treatment. First, elite-school students have

<sup>&</sup>lt;sup>2</sup>Some of the secondary grammar schools offer a 8-years long academic track. Due to data limitations we do not study these 8-years long academic tracks, and drop students who enrolled in these 8-years long academic tracks from all of our samples.

Grade	1–4	5–6	7–8	9–12		
Age	6–9	10-11	12–13	14–17		
Path				Regular track in a secondary grammar school		
	Drimowy achool			Vocational school		
	PIIII	ary school	6-years long academic track in a secondary grammar school (elite scho			
		8-1	years long	g academic track in a secondary grammar school (excluded)		

Table 1: The overview of the education system in Hungary

*Notes:* The table provides an overview of the education system in Hungary. The regular and the 6-years long academic tracks take place in secondary grammar schools. We refer to the 6-years long academic track in a secondary grammar school as an elite school. We exclude the 8-years long academic track from our analysis.

academically stronger peers, which follows from the combination of merit-based admission and the high demand. Second, elite-school students enter the secondary education environment sooner than non-elite-school students. Teachers in secondary schools are typically more qualified than primary school teachers, and they are more experienced with more mature students. Third, elite schools offer a more advanced, higher-paced curriculum. Finally, elite-school students switch school at age 12. By contrast, students who do not enroll in an elite school attend their primary school for an additional 2 years, and switch only at age 14.

# **3** Data and summary statistics

This section describes the data we use to estimate the effect of enrollment in an elite school on elite-school students' academic achievement. We begin, in Section 3.1, by describing the data and discussing the construction of our samples. In Section 3.2, we present summary statistics showing that elite-school students are positively selected based on socioeconomic status and academic achievement, and that the peer quality of elite-school students is better than those of non-elite-school students.

## 3.1 Data

For the analysis, we use administrative data from the National Assessment of Basic Competencies (NABC). Our data are longitudinal and cover every student in grades 6, 8, and 10 in the period of 2008–2014. The backbone of our administrative data are the standardized test scores of the NABC. The NABC is similar to OECD's Programme for International Student Assessment (PISA), but it extends to every student in grades 6, 8, and 10. The NABC (and the corresponding survey on students' background) is conducted annually in the last week of May. The NABC measures students' mathematics and reading skills. Students' 6th-grade test scores are measured on a scale, which has a mean of 1,500 and standard deviation of 200. In grades 8 and 10, the test scores are standardized in a way, such that the test scores are comparable over time (e.g., a student's 6th-grade and 8th-grade test scores are comparable) and across cohorts (e.g., the average test scores in grade 8 are comparable across cohorts). The standardized test scores are of low-stakes, they are graded blindly and externally, and they are not top-coded.<sup>3</sup>

Our data also include information on students' demographics (gender), socioeconomic status (the number of books at home, disadvantaged status, and parental education), and schools (school identifiers in grades 6, 8, and 10, school type, the county of the school, and the type of the settlement where the school is located). We also have rich information on students' academic achievement. Our data also include students' GPA in grade 5, i.e., one year prior the students' first NABC taking place. Finally, our data have information on students' 8th-grade mathematics grade.<sup>4</sup> These grades are given by students' own teachers and they are measured on a scale of 1 to 5.

We study two (overlapping) samples in order to maximize the power of our analysis. When studying short-run outcomes (i.e., outcomes measured 2 years after elite-school enrollment), we focus on students who we observe in grades 6 and 8. We refer to this sample as the *8th-grade sample*. The 8th-grade sample consists of five cohorts, whose 8th-grade outcomes are measured in the period of 2010–2014. When studying medium-run outcomes (i.e., outcomes measured 4 years after elite-school enrollment), we focus on students who we observe in grades 6, 8, and 10. We refer to this sample as the *10th-grade sample*. The 10th-grade sample consists of three cohorts, whose 10th-grade outcomes are measured in the period of 2012–2014.

<sup>&</sup>lt;sup>3</sup>Figure 3 presents the cumulative distribution function of students' 6th-grade mathematics and reading test scores.

<sup>&</sup>lt;sup>4</sup>Information on students' socioeconomic status (number of books at home and parental education) and their school grades (5th-grade GPA and 8th-grade mathematics grade) is self-reported.

In our analysis we make a number of sample restrictions to focus on students for whom enrolling in an elite school is a viable option. First, we focus on students' with a complete academic path with no missing information. Second, our samples include students who are enrolled either in an elite or in a primary school in grade 8. Third, we focus on students whose propensity to enroll in an elite school is non-negligible. Therefore, our samples only include students (1) whose 5th-grade GPA is at least 4, (2) whose 6th-grade mathematics grade is at least 3, and (3) who attend a primary school in grade 6 from which at least one student enrolled in an elite school in our sample period. Appendix Table C1 provides information on the sample size reduction for each sample restriction. After the sample restrictions, we end up with about 25,000 students in each cohort (approximately 25% in each cohort).

In our analysis, we standardize students' test scores on our restricted samples. We conduct our analysis for four student groups, defined by (baseline) ability and gender, separately. We refer to students whose 6th-grade standardized test score is below the sample median as *lowability students*, and whose 6th-grade standardized test score is above the sample median as *high-ability students*.

#### 3.2 Summary statistics

Table 2 presents summary statistics of student characteristics by elite-school enrollment for each of our samples. Panel A focuses on students' pre-treatment characteristics. About 55 percent of the students are female. Approximately 37 percent of the students has maximum 150 books at home, 37 percent has between 150 and 600 books at home, and 27 percent has more than 600 books at home. On average, students' 5th-grade GPA is 4.6, on a scale of 1 to 5.<sup>5</sup> The composition of the 8th-grade and 10th-grade samples are almost identical.

Out of the 126,196 students in the 8th-grade sample, 16,702 students (13.2 percent) enrolled in an elite school. Students who enrolled in an elite school have higher socioeconomic status: about 45 percent of elite-school students has more than 600 books at home compared to 24 percent of those who did not enroll in an elite school. Elite-school students' 6th-grade mathematics (reading) test score is higher than the sample average by 53 (50) percent of a

<sup>&</sup>lt;sup>5</sup>The relatively high average 5th-grade GPA follows from the fact that our samples include students whose 5th-grade GPA is at least 4; see Section 3.1.

standard deviation. By contrast, students who did not enroll in an elite school have a 6thgrade test score that is by 6 percent of a standard deviation lower than the sample average in both subjects. These patterns indicate that elite-school students are positively selected based on their socioeconomic status and (baseline) academic achievement.

Panel B of Table 2 reports summary statistics on students' outcomes. On average, students' 8th-grade mathematics grade is 4.0. Elite-school and non-elite-school students' average 8th-grade mathematics grades are similar. By contrast, elite-school students' 8th-grade standardized test scores are considerably higher than those of non-elite-school students. For example, elite-school students' average 8th-grade mathematics test score is 36 percent of a standard deviation higher the sample average. Non-elite-school students' average 8th-grade mathematics test score is 6 percent of a standard deviation lower than the sample average. Summary statistics of the 10th-grade sample indicate that elite-school students' average 10th-grade mathematics test score is 58 percent of a standard deviation higher the sample average. By contrast, non-elite-school students' average 10th-grade mathematics test score is 8 percent of a standard deviation lower than the sample average. These patterns indicate that the differences in standardized scores between elite-school and non-elite-school students persist up until 4 years after elite-school enrollment.

#### Table 2: Summary statistics

	8	8th-grade sample		10th-grade sample		
	Elite-school students (1)	Non-elite-school students (2)	Total (3)	Elite-school students (4)	Non-elite-school students (5)	Total (6)
A. Pre-treatment characteristics						
Female	0.54 (0.50)	0.56 (0.50)	0.55 (0.50)	0.54 (0.50)	0.56 (0.50)	0.55 (0.50)
Max. 150 books at home	0.19 (0.39)	0.39 (0.49)	0.37 (0.48)	0.17 (0.37)	0.36 (0.48)	0.33 (0.47)
B/w 150 and 600 books at home	0.36 (0.48)	0.37 (0.48)	0.37 (0.48)	0.34 (0.47)	0.37 (0.48)	0.37 (0.48)
More than 600 books at home	0.45 (0.50)	0.24 (0.43)	0.27 (0.44)	0.49 (0.50)	0.27 (0.45)	0.30 (0.46)
5th-grade GPA (1–5)	4.77 (0.28)	4.55 (0.34)	4.58 (0.34)	4.78 (0.28)	4.57 (0.34)	4.59 (0.34)
6th-grade mathematics test score (std.)	0.53 (1.01)	-0.08 (0.97)	0.00 (1.00)	0.56 (1.02)	-0.08 (0.97)	0.00 (1.00)
6th-grade reading test score (std.)	0.50 (0.96)	-0.08 (0.98)	0.00 (1.00)	0.51 (0.96)	-0.07 (0.99)	-0.00 (1.00)
B. Outcomes						
8th-grade mathematics grade (1–5)	4.04 (0.87)	3.98 (0.88)	3.99 (0.88)	(.)	(.)	(.)
8th-grade mathematics test score (std.)	0.36 (1.01)	-0.06 (0.99)	0.00 (1.00)	(.)	(.)	(.)
8th-grade reading test score (std.)	0.42 (0.95)	-0.06 (0.99)	0.00 (1.00)	(.)	(.)	
10th-grade mathematics test score (std.)	(.)	(.)	. (.)	0.58 (0.98)	-0.08 (0.98)	0.00 (1.00)
10th-grade reading test score (std.)	(.)	(.)	(.)	0.53 (0.91)	-0.07 (0.99)	-0.00 (1.00)
Number of students	16,702	109,494	126,196	8,850	63,112	71,962

*Notes:* The table presents the means and standard deviations of student characteristics for each sample. Columns (1) and (4) focus on students who did not enroll in an elite school, columns (2) and (5) focus on students who enrolled in an elite school, and columns (3) and (6) focus on the entire sample.

Figure 1 presents the distribution of students' *peer quality*, which is the leave-out mean of peers' 6th-grade standardized test scores within a school. The figure displays the distribution of peer quality before and after elite-school enrollment, i.e., in grades 6 and 8. We present these distributions for elite-school and non-elite-school students separately, and for students' 6th-grade mathematics and reading test scores.<sup>6</sup>

The left panels of Figure 1 show that the peer quality distribution of elite-school students shifts to the right between grades 6 and 8. The average 6th-grade mathematics test score of elite-school students' peers in grade 6 is 1,530. By contrast, the average 6th-grade mathematics test score of elite-school students' peers in the elite school is 1,671. Thus, elite-school students' peer quality substantially improves after they enroll in an elite school.

The right panels of Figure 1 show that the peer quality distribution of non-elite-school students largely remain unchanged between grades 6 and 8. The average 6th-grade mathematics test score of non-elite-school students' peers in grade 6 is 1,512 and it is 1,501 in grade 8. This small reduction in the peer quality of non-elite school students is consistent with positive selection into elite schools.<sup>7</sup> We also note that elite-school students' peer quality considerably exceeds the the peer quality of non-elite-school students in grade 8. Thus, enrollment in an elite school entails having academically stronger peers.

## 4 Empirical strategies

This section discusses our empirical strategies to identify the effect of enrollment in an elite school on the outcome distribution. We begin, in Section 4.1, by deriving a non-parametric bound on the effect of enrollment in an elite school using a weak stochastic dominance assumption (conditional Monotone Treatment Selection). In Section 4.2, we present evidence supporting the validity of our identifying assumption. Finally, in Section 4.3, we discuss a complementary empirical strategy that builds on the selection on observables assumption.

<sup>&</sup>lt;sup>6</sup>When we compute students' peer quality, we include all students in the calculations. Therefore, in Figure 1, we preserve the original scale of students' 6th-grade standardized test scores; see Section 3.1.

<sup>&</sup>lt;sup>7</sup>In the same context, Schiltz et al. (2019) show that the departure of smart peers to elite schools has a small, negative effect on the academic achievement of students who are left behind.



Figure 1: Peer quality and elite-school enrollment

*Notes:* The figure shows the distributions of students' peer quality in grades 6 and 8. Peer quality is the leave-out mean of peers' 6th-grade standardized test scores (within school). The left (right) panels focus on elite-school (non-elite-school) students. The top (bottom) panels focus on peers' 6th-grade mathematics (reading) test scores.

# 4.1 Non-parametric bounds: conditional MTS throughout the outcome distribution

We are interested in the effect of enrollment in an elite school on elite-school students' outcomes, that is, we focus on the the *average treatment effect on the treated* (ATET). We study the entire distribution of potential outcomes. Thus, the causal effect of interest, denoted by  $\tau(\gamma)$ , is the effect of elite-school enrollment on the probability of obtaining an outcome greater than  $\gamma$  for students who enrolled in an elite school:

$$au(\gamma) = \mathbb{P}\left[Y(1) \ge \gamma | D = 1\right] - \mathbb{P}\left[Y(0) \ge \gamma | D = 1\right],$$

where we denote student *i*'s potential outcome by  $Y_i(D)$  and *D* takes a value of one if the student enrolled in an elite school and zero otherwise.

The causal effect of interest focuses on students who enrolled in an elite school. This

means that we identify the effect not only for students on the margin of admission, as studies that exploit priority-score cutoffs, but also for inframarginal students. Thus, we address whether elite schools academically benefit students who enrolled in them, and not weather the expansion of elite-school capacities benefits marginally admitted students.

The identification problem is that the potential untreated outcome distribution of students who enrolled in an elite school is unobservable. Therefore, we make an assumption on the selection of students into elite schools. Specifically, we impose a weak stochastic dominance condition on the potential untreated outcome distribution of students who enrolled in an elite school.

Assumption 1 (Monotone Treatment Selection, MTS). The distribution of potential untreated outcomes of students who enrolled in an elite school weakly dominates that of those who did not enroll in an elite school:<sup>8</sup>

$$\mathbb{P}\left[Y(0) \ge \gamma | D = 1\right] \ge \mathbb{P}\left[Y(0) \ge \gamma | D = 0\right], \quad \forall \gamma.$$
(MTS)

The MTS assumption implies that the effect of enrollment in an elite school does not exceed the difference between the (observed) outcome distributions of students who enrolled in an elite school and of those who did not enroll:

$$\begin{aligned} \tau(\gamma) &= \mathbb{P}\left[Y(1) \ge \gamma | D = 1\right] - \mathbb{P}\left[Y(0) \ge \gamma | D = 1\right] \\ &\leq \mathbb{P}\left[Y(1) \ge \gamma | D = 1\right] - \mathbb{P}\left[Y(0) \ge \gamma | D = 0\right]. \end{aligned}$$

We further sharpen the upper bound on the effect of enrollment in an elite school by assuming that the MTS assumption holds for certain subgroups of students.

**Assumption 2** (conditional MTS). The distribution of potential untreated outcomes of students who enrolled in an elite school weakly dominates that of those who did not enroll in an elite school conditional on each values of the variable of *X*:

$$\mathbb{P}\left[Y(0) \ge \gamma | D = 1, X\right] \ge \mathbb{P}\left[Y(0) \ge \gamma | D = 0, X\right], \quad \forall \gamma.$$
 (conditional MTS)

<sup>&</sup>lt;sup>8</sup>Our MTS assumption requires stochastic dominance of the potential outcome distributions, and thus is stronger than the MTS assumption originally proposed by Manski and Pepper (2000).

In the analysis, for a given value of  $\gamma$ , we use the conditional MTS assumption to derive bounds on  $\mathbb{P}[\Upsilon(0) \leq \gamma | D = 1, X]$  for each values of *X*. We then combine these bounds to obtain an upper bound on the causal effect of interest:<sup>9</sup>

$$\tau(\gamma) \le \sum_{x \in X} \left( \mathbb{P}\left[ Y(0) \le \gamma | D = 0, X = x \right] - \mathbb{P}\left[ Y(1) \le \gamma | D = 1, X = x \right] \right) \mathbb{P}\left[ D = 1 | X = x \right] \quad \forall \gamma \in \mathbb{N}$$

The upper bound on the effect of enrollment in an elite school equals to the corresponding exact matching estimator (Rubin, 1973). The exact matching estimator identifies the effect of enrollment in an elite school on the treated under the Conditional Independence Assumption (CIA). Instead of maintaining the CIA, we assume that selection into an elite school is positive conditional on *X*, and thus we identify an upper bound on the causal effect of interest.

We estimate the non-parametric bound on  $\tau(\gamma)$  by the corresponding sample means and empirical probabilities. The 95% confidence intervals on the causal effect of interest are based on 1,000 bootstrap replications using the methodology derived by Imbens and Manski (2004).

We combine two pre-treatment variables, students' 5th-grade GPA and the number of books at home, to sharpen the bounds on the effect of enrollment in an elite school. We assume that conditional on students' 5th-grade GPA and the number of books at home the potential untreated outcome distribution of students who enrolled in an elite school weakly dominates of those who did not enroll in an elite school. Since admission to elite schools is merit-based, elite-school students are positively selected. Consistent with the merit-based admission procedure, we assume that positive selection is present even conditional on students' 5th-grade GPA, which is a pre-treatment proxy of students' academic ability, and the number of books at home, which is a proxy of socioeconomic status. The next section presents evidence using pre-treatment outcomes to support the validity of our identifying assumptions.<sup>10</sup>

<sup>&</sup>lt;sup>9</sup>Since we investigate the effect of enrollment in an elite school on the entire outcome distribution, we could derive a *no-assumption* lower bound on  $\tau(\gamma)$ . This lower bound is, by construction, never positive, thus we do not report it.

<sup>&</sup>lt;sup>10</sup>Students' 5th-grade GPA and the number of books at home are likely to be positively related to students' potential outcome distribution, and thus are valid monotone instrumental variables (MIVs) (Manski and Pepper, 2000; de Haan, 2011). We find empirical support for the combination of students' 5th-grade GPA and the number of books at home being valid MIVs. However, consistent with the argument of Richey (2016), the combination of the conditional MTS and MIV assumptions does not tighten our bounds.

## 4.2 Validity check

We next assess the validity of the conditional MTS assumption for the interaction of students' 5th-grade GPA and the number of books at home. A prerequisite of the conditional MTS assumption is a sufficient overlap between elite-school and non-elite-school students for each value of the interaction of students' 5th-grade GPA and the number of books at home (cf. Common Support Assumption). Moreover, if the conditional MTS assumption is met, then the distribution of elite-school students' pre-treatment outcomes should weakly dominate those of non-elite-school students' pre-treatment outcomes. We use students' 6th-grade standardized test score, which is realized prior to elite-school enrollment, to test this implication.

We first provide evidence supporting the overlap between elite-school and non-elite-school students' characteristics. Figure 2 presents the share of students who enrolled in an elite school for each value of the combination of 5h-grade GPA and the number of books at home. The figure splits the sample by (pre-treatment) ability and gender. There is no combination of 5th-grade GPA and the number of books at home such that the share of elite-school students is below 1 percent.<sup>11</sup>

We next provide graphical evidence supporting the MTS assumption. Figure 3 displays the distribution of students' 6th-grade standardized test scores by elite-school enrollment and gender. The figure shows that the 6th-grade test score distribution of students who enrolled in an elite school weakly dominates of those who did not enroll for each gender and test type (mathematics and reading).

Figure 4 presents a formal test of the validity of the conditional MTS assumption. For each value of the combination of students' 5th-grade GPA and the number of books at home, we perform a one-sided Kolgomorov-Smirnov test. Figure 4 presents the corresponding p-values of the Kolgomorov-Smirnov test by gender and pre-treatment ability for each value of the combination of 5th-grade GPA and number of books at home. Out of the 264 tests we perform, we reject the null hypothesis only 6 times at a 10 percent significance level. We interpret these results as strong evidence supporting the validity of the conditional MTS assumption.<sup>12</sup>

<sup>&</sup>lt;sup>11</sup>Appendix Figure A1 provides the same information for the 10th-grade sample, with the same conclusion.

<sup>&</sup>lt;sup>12</sup>Appendix Figure A2 provides evidence supporting the validity of the conditional MTS assumption for the 10th-grade sample.



Figure 2: Validity check: Elite-school enrollment and student characteristics

*Notes:* The figure presents the share of students who enrolled in an elite school by student characteristics. Each cell shows the share of elite-school students for a combination of 5th-grade GPA and the number of books at home. In the top panels (bottom) high-ability/low-ability is defined as having 6th-grade mathematics (reading) test score above/below the median. Sample: 8th-grade sample, N = 126,196.

## 4.3 School value-added

The non-parametric bound approach offers several advantages (e.g., mild identifying assumptions with testable implications), however, its ability to recover informative estimates may be limited. Therefore, we consider a complementary empirical strategy, which builds on selection on observables—a more demanding assumption.

Assumption 3 (Selection on observables).

$$\mathbb{P}\left[Y(d) \ge \gamma | D = d, Z\right] = \alpha_d^{\gamma} + \beta_d^{\gamma} Z, \quad d = 0, 1; \forall \gamma,$$

where Z is a vector of variables.

The selection on observables assumption provides point identification for the causal effect

**Figure 3:** Validity-check: The distribution of students' 6th-grade standardized test scores by elite-school enrollment



Elite-school students — Non-elite-school students

*Notes:* The figure displays the cumulative distribution function of students' 6th-grade standardized test score by elite-school enrollment and gender. The left (right) panels present the distributions for female (male) students. The top (bottom) panels present the distribution of the 6th-grade mathematics (reading) test scores.

of interest:

$$\tau(\gamma) = \sum_{z \in Z} \left( \alpha_1^\gamma - \alpha_0^\gamma + (\beta_1^\gamma - \beta_0^\gamma) z \right) \mathbb{P} \left[ D = 1 | Z = z \right].$$

The selection on observables assumption ensures that an ordinary least squares (OLS) regression of  $Y_i$  on an indicator of elite-school enrollment interacted with  $Z_i$  recovers unbiased estimates of  $\alpha_d^{\gamma}$  and  $\beta_d^{\gamma}$ . We consider two specifications. First, the vector *Z* includes students' 6th-grade standardized test scores and cohort fixed effects (*simple* model). Second, the vector *Z* includes additional covariates, such as 5th-grade GPA, number of books at home, education of the mother, education of the father, being disadvantaged, the county of the school, and the type of the residence where the school is located (*full* model).<sup>13</sup> Both specifications resemble the commonly used "*school value-added*" approach to evaluate the effectiveness of

<sup>&</sup>lt;sup>13</sup>Appendix C describes the construction of our variables, and Appendix Table A1 presents summary statistics of the additional covariates of the school value-added specifications.



Figure 4: Validity-check: The p-values of the Kolgomorov-Smirnov test

*Notes:* The figure displays the p-values of the one-sided Kolgomorov-Smirnov test. The Kolgomorov-Smirnov test tests the equality of the distributions of elite-school and non-elite-school students' 6th-grade standardized test scores. Each cell shows the p-value for a combination of 5th-grade GPA and the number of books at home. In the top panels (bottom) high-ability/low-ability is defined as having 6th-grade mathematics (reading) test score above/below the median. Sample: 8th-grade sample, N = 126,196.

#### schools (Koedel et al., 2015).

The plausibility of the selection on observables assumption is debated in the context of school value-added (school VA) estimates (see e.g., Chetty et al., 2014a,b; Guarino et al., 2015; Rothstein, 2010, 2017). For elite schools in Amsterdam, Oosterbeek et al. (2020) finds that school value-added estimates are severely biased when they are compared to admission lottery-based estimates. In the context of Hungarian elite schools, we view the school value-added estimates complementary to our non-parametric bound approach. We note that when the selection on observables assumption is violated, the direction of the bias of the school value-added estimate is unclear. Therefore, the school value-added estimates might be less informative about our causal effect of interest than the non-parametric bounds.

# 5 Results

This section presents our results. We begin, in Section 5.1, by showing that enrollment in an elite school has a negative effect on the short-run academic achievement of female and low-ability students. In Section 5.2, we show that our non-parametric bounds strategy cannot rule out moderately positive effects on academic achievement 4 years after enrollment. Finally, in Section 5.3, using value-added models, we show that elite schools improve students' academic achievement 4 years after enrollment.

#### 5.1 Short-run academic achievement

We begin by studying the effect of enrollment in an elite school on elite-school students' 8thgrade mathematics grade. Figure 5 displays the upper bounds on the causal effect of interest, split by ability and gender.<sup>14</sup> The figure also displays the raw (unconditional) differences between the outcomes of elite-school and non-elite school students for each group.

Figure 5 shows that the upper bounds are negative throughout the outcome distribution for each gender–ability group. For example, enrollment in an elite school decreases the probability of having an 8th-grade mathematics grade larger than 4.5 by at least 15 percentage points for high-ability male students. The upper bound of 15 percentage points corresponds to a 21 percent relative decrease (Appendix Figure B1). For high-ability students, the upper bounds are the lowest at the top of the distribution. The upper bounds of the causal effect are considerably lower than the unconditional differences, which is consistent with positive selection into elite schools.

The negative upper-bound estimates on students' 8th-grade mathematics grade do not necessarily reflect negative effects on students' skill formation. Alternative explanations are ceiling effects, grading on a curve (Calsamiglia and Loviglio, 2019), or more demanding study requirements in elite schools. To rule out these alternative explanations, we next examine the effect of elite schools on students' 8th-grade standardized test scores. This test score is

<sup>&</sup>lt;sup>14</sup>The figure also displays the 95% confidence intervals on the causal effect of interest. Since we present the upper bound estimates of the causal effect exclusively, we mark the area between the upper confidence band and the estimate itself (shaded area).

not top-coded, blindly graded, and standardized nationwide, therefore, it is not susceptible to the above mentioned alternative explanations.



**Figure 5:** The effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grade

*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grade (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade mathematics test score is below or above the median. Students' 6th-grade mathematics grade is measured on the scale of 1–5. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.

Figure 6 presents the upper bounds on the effect of enrollment in an elite school on the distribution of 8th-grade standardized test scores. The figure displays the upper bounds on the deciles of the distribution of students' mathematics and reading test scores.<sup>15</sup>

The figure shows that enrollment in an elite school has a negative effect on female and low-ability elite-school students' mathematics test scores two years after enrollment. Eliteschool enrollment reduces the probability of scoring above the median in mathematics by

<sup>&</sup>lt;sup>15</sup>To simplify the exposition, Figure 6 displays the bounds only on the bottom- and top-6 deciles of the outcome distribution of low- and high-ability students, respectively.

more than 7.5 percentage points, corresponding to 30 percent, for low-ability female students, and by more than 1 percentage points, corresponding to 10 percent, for low-ability male students. For high-ability female students, we find that elite-school enrollment reduces the probability of scoring in the top 10 percent in mathematics by more than 2 percentage points, corresponding to 10 percent (Appendix Figure B2). The figure also shows that enrollment in an elite school does not meaningfully increase the 8th-grade mathematics test scores of low-ability male students. The upper bounds are actually negative for the fourth and fifth deciles of the outcome distribution. If anything, elite-school enrollment may only raise highachieving male students' mathematics test score at the top of the outcome distribution, by at most 2–5 percentage points.

The bottom panels of the figure show the effect of enrollment in an elite school on students' 8th-grade reading test scores. We find that the effect is negative for low-ability female students. The bounds do not exclude positive effects for male and high-achieving female students.

A potential mechanism underlying the negative short-run effects is that elite-school students switch schools, and it may take time for them to adapt to the new environment. We test this hypothesis by conducting a heterogeneity analysis. To this end, we focus on elite-school students who attend a comprehensive school, and thus enrollment in an elite school does not involve switching schools. Figure 7 presents the estimates for this subsample of students. The upper bounds are negative for both male and female students (irrespective of ability) in mathematics. These results suggest that the fact that students in the comparison group do not switch school does not drive our short-run results.<sup>16</sup>

#### 5.2 Medium-run academic achievement

Having found that enrollment in an elite school has a negative effect on female and lowability students' 8th-grade mathematics test score, we next investigate whether these negative effects persist in grade 10. Figure 8 presents the upper bounds on the effect of enrollment in an elite school on the distribution of 10th-grade test scores. We find that the upper bounds

<sup>&</sup>lt;sup>16</sup>Appendix Figure A3 shows the estimates for elite-school students who did switch to an elite school. The estimates are largely similar to those presented in Figure 6.





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grades (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.

are positive, and relatively large in magnitude, for each subgroup and test (mathematics and reading). These findings are consistent with negative as well as substantial positive effects, and therefore they are inconclusive about the question whether the short-run negative effects persist.

A potential explanation for the high upper bounds is that students who did not enroll in an elite school attend low value-added schools. This concern, for example, is particularly pronounced for vocational schools, whose curriculum has less of an academic orientation. We thus focus on the subset of secondary grammar schools that have elite school tracks and regular tracks as well (see Table 1). Figure 9 presents the upper bounds on this subset of secondary grammar schools. We find that the upper bounds are positive, but considerably lower than the ones presented in Figure 8. For example, the upper bound on the effect of





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample – comprehensive schools, N = 111,501.

having a mathematics test score in the top decile for high-ability male students drops from 12.5 to 6 percentage points. These findings indicate that low value-added secondary-schools (e.g., vocational schools) do not drive the positive upper bounds. Yet, these bounds do not reveal whether the negative effects of elite schools persist.

#### 5.3 School value-added

As our non-parametric bounds strategy is inconclusive about the sign of the effects of eliteschool enrollment in the medium run, we turn to the results of our school value-added models. Figure 10 presents the school value-added estimates for students' 8th-grade standardized test scores, in each gender–ability group using two alternative specifications. The simple





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample, N = 71,962.

model controls for students' 6th-grade standardized test score and cohort fixed effects. The full model adds additional controls for student and school characteristics (see Section 4.3).

The figure corroborates that enrollment in an elite school has a negative effect on female elite-school students' 8th-grade mathematics test scores throughout the outcome distribution. The simple model shows that enrollment in an elite school decreases the probability of having a mathematics test score above the median by 5 percentage points for low-ability, and 3 percentage points for high-ability female students. The full model with additional covariates suggests that the effect is about minus 7.5 percentage points for low-ability, and minus 8 percentage points for high-ability female students. The negative relative effect of elite-school enrollment is larger for low-ability female students and at the top of the outcome distribution (Appendix Figure B6).



Figure 9: The effect of elite-school enrollment on the distribution of elite-school students'



*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample – elite-school subsample, N = 21,384.

Figure 10 also reiterates that elite-school enrollment does not meaningfully increase the 8th-grade mathematics test scores of low-ability male students. The estimates of the simple school VA model are close to zero for both low- and high ability male students throughout the entire outcome distribution. The full model with additional covariates indicates that the effect is negative. Similar to female students, we find that the size of the relative effects are larger for low-ability male students relative to high-ability male students, and at the top of the outcome distribution (Appendix Figure B6).

The school value-added estimates are less robust when we study the effect of elite-school enrollment on elite-school students' 8th-grade reading scores. The simple model yields positive estimates for each ability-gender group throughout the outcome distribution. By contrast, the full model results in negative or insignificant estimates. We find that the full model's results are consistent with our non-parametric bounds (Appendix Figure A4).





*Notes:* The figure presents the school value-added estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores. The figure presents school VA estimates for the deciles of the outcome distribution. The top (bottom) panels focus on mathematics (reading). The left (right) panels focus on students whose 6th-grade test score is above (below) the median. The dashed lines refer to the estimates of the simple school VA model (6th-grade standardized test score, cohort fixed effects) and the solid lines refer to the full school VA model (6th-grade standardized test score, cohort fixed effects, 5th-grade GPA, number of books at home, parental education, disadvantaged status, county of the school, type of the settlement where the school is located). The shaded are represents the 95% confidence intervals around the school VA estimates. The confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.

Figure 11 presents the school VA estimates for elite-school students' 10th-grade standardized test scores. We find that elite-school enrollment has a positive effect on elite-school students' 10th-grade test scores for both ability groups. The full model suggests that eliteschool enrollment increases the probability of having a mathematics test score above the median by 10 percentage points for low-ability female students, and 12.5 percentage points for low-ability male students. For high-ability students, we find that the effect on the probability of having a mathematics test score above the 9th decile is 7 percentage points for girls, and 8 percentage points for boys. The relative effect elite-school enrollment on the mathematics test score is heterogeneous for high-ability students. We find that the relative effect for high-ability female students is higher than those of high-ability male students. By contrast, the relative effects do not differ between low-ability female and male students. We also find that the relative effect is higher at the top of the outcome distribution (Appendix Table B7).





*Notes:* The figure presents the school value-added estimates of the effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores. The figure presents school VA estimates for the deciles of the outcome distribution. The top (bottom) panels focus on mathematics (reading). The left (right) panels focus on students whose 6th-grade test score is below (above) the median. The dashed lines refer to the estimates of the simple school VA model (6th-grade standardized test score, cohort fixed effects) and the solid lines refer to the full school VA model (6th-grade standardized test score, cohort fixed effects, 5th-grade GPA, number of books at home, parental education, disadvantaged status, county of the school, type of the settlement where the school is located). The shaded are represents the 95% confidence intervals around the school VA estimates. The confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample, N = 71,962.

# 6 Conclusions

We have studied the effect on enrollment in an elite school on elite-school students' short- and medium-run academic achievement. Motivated by the fact that local effects identified at the margin of admission may not generalize to the average elite-school student, the causal effect of interest has been the average treatment effect on the treated. To identify the causal effects of interest, we have used non-parametric bounds and school value-added models. To further understand the heterogeneous effects of elite schools, we have conducted our analysis by gender-ability groups, and we have studied the effects throughout the outcome distribution.

Our main finding is that enrollment in an elite school has a negative effect on female and low-ability students' mathematics test scores on the short run. By contrast, our nonparametric bounds strategy does not allow us to exclude small, positive effects for highability male students. We argue that the short-run negative effects are not the consequence of grading policies (e.g., grading on a curve, ceiling effects), but they reflect real effects in skill formation. As non-parametric bounds are uninformative about the sign of the mediumrun effects, we have used school value-added models to test whether the negative short-run effects persist. These school value-added estimates suggest that elite-school enrollment has a positive effect on academic achievement for each gender-ability group on the medium run.

The most salient difference between an average elite-school student and a student on the margin of admission is her (baseline) academic ability. Therefore, our analysis has focused on effect heterogeneity along the ability distribution. Splitting the sample by baseline ability suggests that high-ability students benefit more from elite-school enrollment. By studying the effect of elite-school enrollment throughout the outcome distribution, our findings indicate that the benefits of elite schools are concentrated at the top of the test score distribution.

Our study design does not allow us to identify the exact mechanisms driving our results. Nonetheless, heterogeneity analysis along school characteristics enables us to exclude some important mechanisms. First, the negative upper bounds on the short-run effects in comprehensive schools suggest that school switching does not explain the negative short-run effects of elite schools. Second, the positive upper bounds on medium-run outcomes are not solely driven by less selective schools in the counterfactual.

We have focused on how elite-school enrollment affect academic achievement. However,

parents may also value schools along other dimensions, such as non-cognitive skills, field of study, college enrollment, or labor market outcomes (see, e.g., Beuermann and Jackson, Forthcoming; Beuermann et al., 2018).<sup>17</sup> We view the understanding of how elite schools affect these outcomes as fruitful directions for future research.

<sup>&</sup>lt;sup>17</sup>The strong positive association between Hungarian students' standardized test scores and their labor market outcomes (such as employment and wages) suggests that the medium-run benefits in academic achievement may turn into long-run gains in the labor market (Hermann et al., 2019).

# References

- Abdulkadiroglu, Atila, Joshua Angrist, and Parag Pathak. 2014. "The Elite Illusion: Achievement Effects at Boston and New York Exam Schools." *Econometrica*, 82(1): 137–196.
- Anderson, Kathryn, Xue Gong, Kai Hong, and Xi Zhang. 2016. "Do Selective High Schools Improve Student Achievement? Effects of Exam Schools in China." *China Economic Review*, 40 121–134.
- **Barrow, Lisa, Lauren Sartain, and M. De la Torre.** Forthcoming. "Increasing Access to Selective High Schools through Place-based Affirmative Action: Unintended Consequences." *American Economic Journal: Applied Economics.*
- **Beuermann, Diether W., and C. Kirabo Jackson.** Forthcoming. "The Short and Long-Run Effects of Attending the Schools that Parents Prefer." *Journal of Human Resources*.
- Beuermann, Diether W., C. Kirabo Jackson, Laia Navarro-Sola, and Francisco Pardo. 2018. "What is a Good School, and Can Parents Tell? Evidence on the Multidimensionality of School Output." Working Paper 25342, National Bureau of Economic Research.
- Bui, Sa A., Steven G. Craig, and Scott A. Imberman. 2014. "Is Gifted Education a Bright Idea? Assessing the Impact of Gifted and Talented Programs on Students." *American Economic Journal: Economic Policy*, 6(3): 30–62.
- **Calsamiglia, Caterina, and Annalisa Loviglio.** 2019. "Grading on a Curve: When Having Good Peers is not Good." *Economics of Education Review*, 73(101916): 1–21.
- Chetty, Raj, John N. Friedman, and Jonah E. Rockoff. 2014a. "Measuring the Impacts of Teachers I: Evaluating Bias in Teacher Value-Added Estimates." *American Economic Review*, 104(9): 2593–2632.
- Chetty, Raj, John N. Friedman, and Jonah E. Rockoff. 2014b. "Measuring the Impacts of Teachers II: Teacher Value-Added and Student Outcomes in Adulthood." *American Economic Review*, 104(9): 2633–2679.

- Clark, Damon. 2010. "Selective Schools and Academic Achievement." *The B.E. Journal of Economic Analysis & Policy*, 10(1): 1–40.
- **Cullen, Julie Berry, Brian A. Jacob, and Steven Levitt.** 2006. "The Effect of School Choice on Participants: Evidence from Randomized Lotteries." *Econometrica*, 74(5): 1191–1230.
- **Deming, David J.** 2011. "Better Schools, Less Crime?" *Quarterly Journal of Economics*, 126(4): 2063–2115.
- **Dobbie, Will, and Jr. Fryer, Roland G.** 2011. "Are High-Quality Schools Enough to Increase Achievement Among the Poor? Evidence from the Harlem Children's Zone." *American Economic Journal: Applied Economics.*, 3(3): 158–187.
- **Dobbie, Will, and Jr. Fryer, Roland G.** 2014. "The Impact of Attending a School with High-Achieving Peers: Evidence from the New York City Exam Schools." *American Economic Journal: Applied Economics*, 6(3): 58–75.
- **Guarino, Cassandra M., Mark D. Reckase, and Jeffrey M. Wooldridge.** 2015. "Can Value-Added Measures of Teacher Performance Be Trusted?" *Education Finance and Policy*, 10(1): 117–156.
- de Haan, Monique. 2011. "The Effect of Parents' Schooling on Child's Schooling: A Nonparametric Bounds Analysis." *Journal of Labor Economics*, 29(4): 859–892.
- de Haan, Monique, and Edwin Leuven. 2020. "Head Start and the Distribution of Long-Term Education and Labor Market Outcomes." *Journal of Labor Economics*, 38(3): 727– 765.
- Hermann, Zoltán, Dániel Horn, János Köllő, Anna Sebők, András Semjén, and Júlia Varga. 2019. "Szövegértési és matematikai kompetencia hatása a keresetre és a foglalkoztatási esélyekre." In *Munkaerőpiaci Tükör 2018*. eds. by Károly Fazekas, Márton Csillag, Zoltán Hermann, and Ágota Scharle, Budapest: Közgazdaság- és Regionális Tudományi Kutatóközpont, 45–52.

- Horn, Dániel. 2013. "Diverging Performances: The Detrimental Effects of Early Educational Selection on Equality of Opportunity in Hungary." *Research in Social Stratification and Mobility*, 32(June): 25–43.
- **Imbens, Guido W., and Charles F. Manski.** 2004. "Confidence Intervals for Partially Identified Parameters." *Econometrica*, 72(6): 1845–1857.
- Jackson, Kirabo C. 2010. "Do Students Benefit from Attending Better Schools? Evidence from Rule-Based Student Assignments in Trinidad and Tobago." *The Economic Journal*, 120(549): 1399–1429.
- Koedel, Cory, Kata Mihaly, and Jonah E. Rockoff. 2015. "Value-Added Modeling: A Review." *Economics of Education Review*, 47 180–195.
- Lucas, Adrienne M., and Isaac M. Mbiti. 2014. "Effects of School Quality on Student Achievement: Discontinuity Evidence from Kenya." *American Economic Journal: Applied Economics*, 6(3): 234–63.
- Manski, Charles F., and John V. Pepper. 2000. "Monotone Instrumental Variables, with an Application to the Returns to Schooling." *Econometrica*, 68(4): 997–1012.
- **Oosterbeek, Hessel, Nienke Ruis, and Inge Wolf.** 2020. "Using Admission Lotteries to Estimate the Heterogeneous Effects of Elite Schools." Tinbergen Institute Discussion Paper, 2020-018/V.
- **Pop-Eleches, Cristian, and Miguel Urquiola.** 2013. "Going to a Better School: Effects and Behavioral Responses." *American Economic Review*, 103(4): 1289–1324.
- Richey, Jeremiah. 2016. "An Odd Couple: Monotone Instrumental Variables and Binary Treatments." *Econometric Reviews*, 35(6): 1099–1110.
- **Rothstein, Jesse.** 2010. "Teacher Quality in Educational Production: Tracking, Decay, and Student Achievement." *The Quarterly Journal of Economics*, 125(1): 175–214.
- Rothstein, Jesse. 2017. "Measuring the Impacts of Teachers: Comment." *American Economic Review*, 107(6): 1656–84.

- Rubin, Donald B. 1973. "Matching to Remove Bias in Observational Studies." *Biometrics*, 29(1): 159–183.
- Schiltz, Fritz, Deni Mazrekaj, Daniel Horn, and Kristof De Witte. 2019. "Does It Matter When your Smartest Peers Leave your Class? Evidence from Hungary." *Labour Economics*, 59 79–91, Special Issue on European Association of Labour Economists, 30th annual conference, Lyon, France, 13-15 September 2018.

# A Additional Tables

Section A.1 presents additional summary statistics on the variables we use in our school value-added models. Section A.2 presents validity checks for the 10th-grade sample. Section A.3 presents additional results and compares the non-parametric bounds to the school value-added estimates.

## A.1 Data

	8th-grade sample			10th-grade sample			
	Elite-school students	Non-elite-school students	Total	Elite-school students	Non-elite-school students	Total	
	(1)	(2)	(3)	(4)	(5)	(6)	
A. Pre-treatment student characteri	stics						
Primary education (father)	0.02	0.06	0.05	0.01	0.05	0.04	
•	(0.12)	(0.23)	(0.22)	(0.11)	(0.21)	(0.20)	
Secondary education (father)	0.48	0.69	0.67	0.48	0.69	0.66	
-	(0.50)	(0.46)	(0.47)	(0.50)	(0.46)	(0.47)	
Tertiary education (father)	0.50	0.25	0.28	0.51	0.26	0.29	
-	(0.50)	(0.43)	(0.45)	(0.50)	(0.44)	(0.46)	
Primary education (mother)	0.02	0.07	0.06	0.02	0.06	0.05	
	(0.12)	(0.25)	(0.23)	(0.13)	(0.23)	(0.22)	
Secondary education (mother)	0.40	0.62	0.59	0.39	0.61	0.58	
	(0.49)	(0.49)	(0.49)	(0.49)	(0.49)	(0.49)	
Tertiary education (mother)	0.58	0.32	0.35	0.59	0.34	0.37	
	(0.49)	(0.47)	(0.48)	(0.49)	(0.47)	(0.48)	
Disadvantaged	0.00	0.03	0.03	0.00	0.01	0.01	
	(0.07)	(0.17)	(0.16)	(0.06)	(0.12)	(0.12)	
B. School location							
Capital or county capital	0.59	0.41	0.44	0.59	0.64	0.63	
1 5 1	(0.49)	(0.49)	(0.50)	(0.49)	(0.48)	(0.48)	
Town	0.41	0.40	0.40	0.41	0.36	0.36	
	(0.49)	(0.49)	(0.49)	(0.49)	(0.48)	(0.48)	
Village	0.00	0.19	0.16	0.00	0.00	0.00	
C	(0.00)	(0.39)	(0.37)	(0.00)	(0.07)	(0.06)	
Number of students	16,702	109,494	126,196	8,850	63,112	71,962	

#### Table A1: Additional summary statistics

*Notes:* The table presents the means and standard deviations of student characteristics for each sample. Columns (1) and (4) focus on students who did not enroll in an elite school, columns (2) and (5) focus on students who enrolled in an elite school, and columns (3) and (6) focus on the entire sample.

35

## A.2 Validity check



**Figure A1:** Validity check: Elite-school enrollment and student characteristics – 10th-grade sample

*Notes:* The figure presents the share of students who enrolled in an elite school by student characteristics. Each cell shows the share of elite-school students for a combination of 5th-grade GPA and the number of books at home. In the top panels (bottom) high-ability/low-ability is defined as having 6th-grade mathematics (reading) test score above/below the median. Sample: 10th-grade sample, N = 71,962.



**Figure A2:** Validity check: The p-values of the Kolgomorov-Smirnov test – 10th-grade sample

*Notes:* The figure displays the p-values of the one-sided Kolgomorov-Smirnov test. The Kolgomorov-Smirnov test tests the equality of the distributions of elite-school and non-elite-school students' 6th-grade standardized test scores. Each cell shows the p-value for a combination of 5th-grade GPA and the number of books at home. In the top panels (bottom) high-ability/low-ability is defined as having 6th-grade mathematics (reading) test score above/below the median. Sample: 10th-grade sample, N = 71,962.

## A.3 Results

**Figure A3:** The effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores: Non-comprehensive schools



*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample – non-comprehensive schools, N = 124,189.





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores (solid lines) along with the school value-added estimates of full model (dashed lines). The figure presents the estimates for the deciles of the outcome distribution. We report the estimates separately by gender and ability. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the 95% confidence intervals (for the bound estimate, only the upper confidence bound is plotted) based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores (solid lines) along with the school value-added estimates of full model (dashed lines). The figure presents the estimates for the deciles of the outcome distribution. We report the estimates separately by gender and ability. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the 95% confidence intervals (for the bound estimate, only the upper confidence bound is plotted) based on 1,000 bootstrap draws. Sample: 10th-grade sample, N = 71,962.

# **B** The relative effects of elite-school enrollment

This Appendix presents non-parametric bounds on the *relative ATET*, i.e., the relative effect of enrollment in an elite school for elite-school students. We define the relative ATET as follows:

$$\text{Relative ATET} = \frac{\mathbb{P}\left[Y(1) > \gamma | D = 1\right]}{\mathbb{P}\left[Y(0) > \gamma | D = 1\right]} - 1 = \frac{\tau(\gamma)}{\mathbb{P}\left[Y(1) > \gamma | D = 1\right] - \tau(\gamma)}, \quad \forall \gamma$$

In this Appendix, we present the relative ATET estimates for each figure presented in the main text.





*Notes:* The figure presents our upper-bound estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grade (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade mathematics test score is below or above the median. Students' 6th-grade mathematics grade is measured on the scale of 1–5. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.

**Figure B2:** The relative effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores



*Notes:* The figure presents our upper-bound estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grades (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.





*Notes:* The figure presents our upper-bound estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 8th-grade mathematics grades (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample – comprehensive schools, N = 111,501.

**Figure B4:** The relative effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores



*Notes:* The figure presents our upper-bound estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample, N = 71,962.





*Notes:* The figure presents our upper-bound estimates of the effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores (solid lines). The dashed line denotes the raw difference between the outcomes of elite-school and non-elite-school students. We report the estimates separately by gender and low- and high-ability students. Low- and high-ability students are defined by whether the students' 6th-grade standardized test score is below or above the median. The shaded area represents the area between the upper confidence band (95%) and the upper bound estimate itself. The 95% confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample – elite secondary grammar schools, N = 21,384.





*Notes:* The figure presents the school value-added estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores. The figure presents school VA estimates for the deciles of the outcome distribution. The top (bottom) panels focus on mathematics (reading). The left (right) panels focus on students whose 6th-grade test score is above (below) the median. The dashed lines refer to the estimates of the simple school VA model (6th-grade standardized test score, cohort fixed effects) and the solid lines refer to the full school VA model (6th-grade standardized test score, cohort fixed effects, 5thgrade GPA, number of books at home, parental education, disadvantaged status, county of the school, type of the settlement where the school is located). The shaded are represents the 95% confidence intervals around the school VA estimates. The confidence intervals are based on 1,000 bootstrap draws. Sample: 8th-grade sample, N = 126,196.

**Figure B7:** The relative effect of elite-school enrollment on the distribution of elite-school students' 10th-grade standardized test scores: School VA



*Notes:* The figure presents the school value-added estimates of the **relative** effect of elite-school enrollment on the distribution of elite-school students' 8th-grade standardized test scores. The figure presents school VA estimates for the deciles of the outcome distribution. The top (bottom) panels focus on mathematics (reading). The left (right) panels focus on students whose 6th-grade test score is above (below) the median. The dashed lines refer to the estimates of the simple school VA model (6th-grade standardized test score, cohort fixed effects) and the solid lines refer to the full school VA model (6th-grade standardized test score, cohort fixed effects, 5th-grade GPA, number of books at home, parental education, disadvantaged status, county of the school, type of the settlement where the school is located). The shaded are represents the 95% confidence intervals around the school VA estimates. The confidence intervals are based on 1,000 bootstrap draws. Sample: 10th-grade sample, N = 71,962.

# C Data (for online publication)

This Appendix describes our data. We begin, in Section C.1, by describing our sample restrictions. In Section **??**, we explain how we construct our variables.

## C.1 Sample restrictions

Table C1 presents the evolution of the sample size as a result of our sample restrictions.

2010	2011	2012	2013	2014	Total
104,266	96,843	92,966	89,913	87,542	471,530
96,212	89,005	85,245	81,919	80,065	432,446
76,875	70,777	68,278	66,636	65,651	348,217
91,372	84,562	81,441	75,971	74,637	407,983
62,637	57,651	56,943	53,660	53,703	284,594
27,328	25,550	25,326	24,275	23,717	126,196
		102,037	95,649	90,188	287,874
		90,315	83,554	78,727	252,596
		72,697	68,041	64,963	205,701
		84,911	78,561	74,433	237,905
		54,062	48,945	47,365	150,372
		25,371	23,519	23,072	71,962
	2010 104,266 96,212 76,875 91,372 62,637 27,328	20102011104,26696,84396,21289,00576,87570,77791,37284,56262,63757,65127,32825,550	201020112012104,26696,84392,96696,21289,00585,24576,87570,77768,27891,37284,56281,44162,63757,65156,94327,32825,55025,326102,03790,31572,69784,91154,06225,371	2010201120122013104,26696,84392,96689,91396,21289,00585,24581,91976,87570,77768,27866,63691,37284,56281,44175,97162,63757,65156,94353,66027,32825,55025,32624,275I02,03795,64990,31583,55472,69768,04184,91178,56154,06248,94525,37123,51925,37123,519	20102011201220132014104,26696,84392,96689,91387,54296,21289,00585,24581,91980,06576,87570,77768,27866,63665,65191,37284,56281,44175,97174,63762,63757,65156,94353,66053,70327,32825,55025,32624,27523,717102,03795,64990,18890,31583,55478,72772,69768,04164,96384,91178,56174,43354,06248,94547,36525,37123,51923,072

Table C1: Evolution of the sample size

*Notes:* The first row shows the number of students in our raw data, the National Assessment of Basic Competencies (NABC), in each of the relevant years and grades. We document how much of them we lose due to missing test scores and missing background variables. We win back a part of this loss by imputing background variables (see Appendix C.3 for more detail), resulting in a sample of more than 80% of the whole cohorts. Unfortunately, we can only link 60-70% of these students to their 6th-grade results. Restricting the sample to those for whom elite school seems to be a relevant option (having good grades and being in a school in 6th grade from which at least one student goes into an elite school in our sample period) further decreases the size: we end up with about 25% of the cohorts. Our samples are highly selective, but that makes them more relevant for our question.

## C.2 Variable description

This Appendix describes the construction of variables in Tables 2 and A1.

- Number of books at home is categorical variable with three values: max. 150 books, between 150 and 600 books, and more than 600 books;
- Mother's education is a categorical variable with three values: primary, secondary, and tertiary education;
- Father's education is a categorical variable with three values: primary, secondary, and tertiary education;
- **Disadvantaged status** is a binary variable, which takes a value of one when a student has a disadvantaged status (i.e., comes from a low-income family);
- The type of the settlement where the school is located is a categorical variable with four values: village, town, county capital, capital;
- **County** of the school is a categorical variable with 20 values;

## C.3 Imputation

Student's characteristics (number of books at home, mother's education, father's education, disadvantaged status) are gained from an extensive background survey that complements the NABC and which the students fill out together with their parents on a voluntary basis (the average completing rate is around 75 percent, see Table C1). As these characteristics should be mainly constant over time, we could exploit the longitudinal aspect of our data to fill out missing values in one year from the corresponding questionnaire of another year. The imputation is done by following a before-after approach: in the 8th-grade sample, we first look for a value in the 6th-grade questionnaire, or if that is still missing, we rely on the 10th-grade questionnaire; in the 10th-grade sample, we impute from grade 8, or if that is still missing, we go on to grade 6.